Author's response to reviews

Title: Evaluation of the Rapid Scale-up of Collaborative TB/HIV Activities in TB Facilities in Rwanda, 2005-2009

Authors:

Eric S Pevzner (ecp9@cdc.gov)
Vandebriel Greet (gv2124@columbia.edu)
Lowrance David (LowranceD@rw.cdc.gov)
Gasana Michel (michelgasana@yahoo.fr)
Finlay Alyssa (avf0@cdc.gov)

Version: 4 Date: 1 June 2011

Author's response to reviews:

June 1, 2011

BMC Editorial Staff,

We have incorporated Dr. Reid’s suggested revision and uploaded the revised track changes version of the manuscript. We are thankful for the continued input from the reviewers and now hope that the manuscript merits publication from BMC.

Respectfully,

Eric Pevzner
U.S. Centers for Disease Control and Prevention

Email: ecp9@cdc
Phone: 404-639-6094